Author's response to reviews

Title: Exercise and Nutrition for Head and Neck Cancer Patients: A Patient Oriented, Clinic-Supported Randomized Controlled Trial

Authors:

Lauren C Capozzi (lcapozzi@ucalgary.ca)
Harold Lau (Harold.Lau2@albertahealthservices.ca)
Raylene A Reimer (reimer@ucalgary.ca)
Margaret McNeely (mmcneely@ualberta.ca)
Janine Giese-Davis (jgieseda@ucalgary.ca)
Nicole Culos-Reed (nculosre@ucalgary.ca)

Version: 2 Date: 12 September 2012

Author's response to reviews: see over
September 12, 2012

BioMed Central Cancer

To Whom It May Concern:

Please accept for submission with revisions our manuscript entitled “Exercise and Nutrition for Head and Neck Cancer Patients: A Patient Oriented, Clinic-Supported Randomized Controlled Trial”.

In follow up to your recommendations, we (1) indicated in the Methods section that “research ethics approval to conduct this study was obtained through the Conjoint Health Research Ethics Board of the Faculties of Medicine, Nursing and Kinesiology, at the University of Calgary (Ethics ID: E-24446),” and that (2) “participants will complete a written, informed consent before participation in the study.”

Again, thank you for your feedback and we look forward to hearing back from the editorial office and publishing.

Sincerely,

Lauren Capozzi on behalf of all authors.
PhD Student, Culos-Reed Health and Wellness Laboratory
Email: lcapozzi@ucalgary.ca